Author's response to reviews

Title: Psychostimulant prescribing trends in a paediatric population in Ireland: A national cohort study

Authors:

Fiona Boland (fionaboland@rcsi.ie)
Udo Reulbach (reulbacu@tcd.ie)
Rose Galvin (rosegalvin@rcsi.ie)
Nicola Motterlini (rosegalvin@rcsi.ie)
Dervla Kelly (dkelly23@tcd.ie)
Kathleen Bennett (BENNETTK@tcd.ie)
Tom Fahey (tomfahey@rcsi.ie)

Version: 2
Date: 26 February 2015

Author's response to reviews:

Dear Editorial Team,

Apologies for the oversight regarding the manuscript.

The revised version contains a title page, line numbers throughout and the 'Introduction' has been changed to 'Background'.

Many thanks and best wishes,

Rose